Author’s response to reviews

Title: Iodine Content of Dietary Salt at Household level and Associated Factors Using Iodometric Titration Methods in Dera District, Northwest Ethiopia

Authors:

Zelalem Alamrew Anteneh (kzolam@gmail.com)
Melese Engdayehu (mengidayehu@gmail.com)
Gedefaw Abeje (abejegedefaw@gmail.com)

Version: 3 Date: 28 Aug 2017

Author’s response to reviews:

Title of the article: “Iodine Content of Dietary Salt at Household level and Associated Factors Using Iodometric Titration Methods in Dera District, Northwest Ethiopia” (ID: NUTN-D-17-00108R1)

Dear editor,

I like to say thank you for your continuous support and very essential revision comments. We made the following changes based on your comments:

1. We corrected the typos in the figure
2. The figure legend was included at the end of the manuscript (line 386)
3. We made some language corrections